Author's response to reviews

Title: The Effect of Bisphosphonates on gene expression: GAPDH as a housekeeping or a new target gene?

Authors:

Maria Teresa Valenti (valenti.lab.oncnoale@inwind.it)
Francesco Bertoldo (francesco.bertoldo@univr.it)
Luca Dalle Carbonare (luca.dallecarbonare@univr.it)
Giuseppe Azzarello (giuseppe.azzarello@libero.it)
Sonia Zenari (soniazenari@univr.it)
Mirko Zanatta (zanatta@univr.it)
Elena Balducci (e.balducci@libero.it)
Orazio Vinante (orazio.vine@libero.it)
Vincenzo Lo Cascio (vincenzo.locasco@univr.it)

Version: 2 Date: 22 October 2005

Author's response to reviews: see over
To
Alex J Pemberton, PhD
Assistant Editor of BMC-series journals

Dear Alex,

please find enclosed the paper entitled “The Effect of Bisphosphonates on gene expression: GAPDH as a housekeeping or a new target gene?”, submitted for a possible publication in BMC Cancer. We revised the paper according to your suggestions. In particular:

- We formatted the abstract with the requested sections and I updated the abstract details in the submission page.
- We changed the manuscript following the format requirements.
- We included a “competing interests” sentence at the end of the manuscript.

We hope that you can find this paper suitable for publication in BMC Cancer.

Thank you for your kind cooperation.

Best wishes,

Luca

Luca Dalle Carbonare, M.D.
Department of Biomedical and Surgical Sciences
Medicina Interna D
University of Verona
Policlinico G.B. Rossi
Piazzale Scuro
37134 Verona - Italy
E-mail: luca.dallecarbonare@univr.it
Tel. +39-045-8074684
Fax +39-045-583041